

Author's response to reviews

Title: Ipsiversive ictal eye deviation in inferioposterior temporal lobe epilepsy—Two SEEG cases report

Authors:

Wei Zhang (zw810501@sina.com)

Xingzhou Liu (sinclairliu@sina.cn)

Lijun Zuo (zuolijun1987@126.com)

Qiang Guo (guoqiang999brain@163.com)

Qi Chen (qi.chen27@gmail.com)

Yongjun Wang (yongjunwang1962@gmail.com)

Version: 3 **Date:** 20 Jan 2017

Author's response to reviews:

I think the figures 1-11 should be a part of the main manuscript. Thank you!